Author’s response to reviews

Title: Acute reduction of serum 8-iso-PGF2-alpha and advanced oxidation protein products in vivo by a polyphenol-rich beverage; a pilot clinical study with phytochemical and in vitro antioxidant characterization.

Authors:

Boris V Nemzer (bnemzer@vandrunen.com)
Liliana C Rodriguez (lilianna@nutraclinicalinc.com)
Linda Hammond (lhammond@ymail.com)
Robert DiSilvestro (rdisilvestro@ehe.osu.edu)
John M Hunter (jhunter@vandrunen.com)
Zbigniew Pietrzkowski (zpietrzkowski@vandrunen.com)

Version: 3 Date: 10 June 2011

Author’s response to reviews:

Dear Editorial Production Team,

We are very pleased that our manuscript has been accepted for publication in Nutrition Journal. We have reviewed the final proof, and made the changes that were suggested by the editorial team. These changes include the following:

• The word “a” was added to the title running head, such that it now reads “Reduction of oxidation markers by a polyphenol-rich beverage”.

• All colored and underlined text was removed as requested.

• Abstract—the word “for” was replaced with the word “in” in the first sentence of the conclusion section.

• Section 3.2 results—“Figures 1–5” was changed to “Figures 1, 2, 3, 4 and 5” as requested.

• Section 4.0 discussion—the word “factor” was changed to the word “role”

• Declaration of competing interests: competing interests for all authors were added as requested.

• Author’s contributions: we put this into paragraph format, as requested

Please let us know if there is anything more that we can do with regard to this manuscript, and we very much look forward to the publication.

Best Regards

Boris Nemzer